Author's response to reviews

Title: Delayed diagnosis of spontaneous extrusion of a renal calculus in an adult male with spina bifida and paraplegia: case report. Measures to be taken to reduce medical errors in spinal cord injury patients

Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Bhakul M Soni (Bhakul.Soni@mail.soh-tr.nwest.nhs.uk)
Gurpreet Singh (gurpreet.singh@ukgateway.net)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradip.Sett@mail.soh-tr.nwest.nhs.uk)

Version: 7 Date: 8 Dec 2001

Dear Editor
Two more illustrations (Figures 7 and 8) are included to highlight the difficulties encountered in making a correct diagnosis of a clinical disorder in spinal cord injury patients because classical symptoms are usually absent in these patients.
Many thanks
Dr S Vaidyanathan